Title: Atopic dermatitis, cutaneous steroids and cataracts in children: a case series.

Authors:

Andrew Tatham (andrewtatham@yahoo.co.uk)

Version: 5 Date: 28 February 2008

Author's response to reviews: see over
28th February 2008

Dear Editor,

Please find attached a revised manuscript for our case series of atopic dermatitis and cataract in children. In the revised manuscript we have shortened the discussion section from 673 to 488 words as recommended. The other sections of the report remain unchanged.

Yours faithfully

Andrew Tatham